Author's response to reviews

Title: vCJD risk in The Republic of Ireland

Authors:

Dr Michael S Harney (michaelsharney@eircom.net)
Dr Azra C Ghani (a.ghani@ic.ac.uk)
Christl A Donnelly (c.donnelly@ic.ac.uk)
Rory McConn Walsh (rmw@rcsi.ie)
Michael Walsh (mwalsh@rcsi.ie)
Rachel Howley (rhowley@rcsi.ie)
Francesca Brett (fbrett@rcsi.ie)
Michael Farrell (mfarrell@indigo.ie)

Version: 7 Date: 18 Nov 2003

Acknowledgements have been inserted.
A typo has been corrected in the methods section under 'travel to UK'. This had said "It was calculated therefore that 2.5% of the exposed Irish population (112,500) were present in the UK at any given time, and therefore were exposed to the UK risk. This represents approximately 2.5% of the UK risk or 1082 'animal equivalents'. This last line should have read "This represents 0.16% of the UK risk (minus 20% exports) or 1082 'animal equivalents'."